Author's response to reviews

Title: Aging Syndrome Genes and Premature Coronary Artery Disease

Authors:

Adrian F Low (mdclowa@nus.edu.sg)
Christopher J O'Donnell (codonnell@partners.org)
Sekar Kathiresan (skathiresan@partners.org)
Brendan Everett (beverett@partners.org)
Claudia U Chae (cchae@partners.org)
Stanley Y Shaw (sshaw@partners.org)
Patrick T Ellinor (pellinor@partners.org)
Calum A MacRae (cmacrae@partners.org)

Version: Date: 28 September 2005

Author's response to reviews:

Dear Sir/Madam,

We would like to submit this revised manuscript entitled "Aging Syndrome Genes and Premature Coronary Artery Disease" as requested for publication in BMC Medical Genetics. We have improved the manuscript as suggested by the reviewers. In particular we have addressed the following aspects;
The reference citations were incorrectly formatted in Endnote and now have been reformatted to conform to the journal style.
All decimals in the manuscript have been reported to the appropriate level of significance.
Finally we have reevaluated the entire manuscript against the Journal's formatting checklist as requested.

All the authors have contributed to the research as described and have read and approved the final manuscript. None of the authors have any conflicts of interest to declare. Please address all correspondence to Dr. MacRae. We hope that you will find this paper to be of interest to your readership.

Thank you for your consideration of this manuscript.

Sincerely,

Calum MacRae on behalf of all the authors.